Title: Assessment of vitamin D and its association with cardiovascular disease risk factors in an adult migrant population: An audit of patient records at a Community Health Centre in Kensington, Melbourne, Australia

Authors:

Thilanga Ruwanpathirana Dr (thilanga.ruwanpathirana@monash.edu)
Christopher M Reid Prof. (chris.reid@monash.edu)
Alice Owen Dr (alice.owen@monash.edu)
David PS Fong Dr (david.fong@doutta.org.au)
Usha Gowda Mrs (usha.gowda@monash.edu)
Andre MN Renzaho A/Prof (andre.renzaho@monash.edu)

Version: 3
Date: 29 August 2014

Author’s response to reviews: see over
Dear Editor-in-chief,

We have re-submitted the manuscript which has been revised according to the directions provided by the reviewers and the editorial board. Amendments made to the manuscript are outlined below.

Kind regards

Professor Christopher Reid

Revisions

Q. * Reviewer 2, Q7: Please describe the criteria for vitamin D deficiency in the methods section as well. It cannot be assumed that the definition stated in the background is the definition that was used for the research.

A. Criteria has been added. Page 7 line 167

Q * Reviewer 2, minor revisions, Q5: The paragraph heading "Association of vitamin D with CHD risk factors and FRS" has not been amended as requested, namely to replace "vitamin D" with "zone of origin".

A. Amended

Q. The manuscript still needs to be copyedited for English usage. There are several grammar errors in the manuscript.

A. The manuscript has been copyedited and the grammar errors were corrected.